Author's response to reviews

Title: Transforming growth factor beta receptor 1 is a new candidate prognostic biomarker after acute myocardial infarction

Authors:

Yvan Devaux (yvan.devaux@crp-sante.lu)
Melanie Bousquenaud (melanie.bousquenaud@crp-sante.lu)
Sophie Rodius (sophie.rodius@crp-sante.lu)
Pierre-Yves Marie (py.marie@chu-nancy.fr)
Fatiha Maskali (f.maskali@genclis.com)
Lu Zhang (lu.zhang@crp-sante.lu)
Francisco Azuaje (francisco.azuaje@crp-sante.lu)
Daniel R Wagner (wagner.daniel@chl.lu)

Version: 4 Date: 1 December 2011

Author's response to reviews:

The following formatting changes have been performed:

1. Colors in Tables 4 and 5 have been replaced by bold characters.
2. Legends to Tables 4 and 5 have been rephrased.
3. Line numbers have been removed.
4. Highlighting/tracking has been removed.
5. The full citation details have been provided for reference 38.
6. Figures numbers have been removed from all figure files.